

**Author’s response to reviews**

**Title:** Traditional healers’ role in the detection of active TB cases in a pastoralist community in Ethiopia: a pilot interventional study

**Authors:**

Bezawit Sima (abemnow@gmail.com; b.t.sima@studemed.uio.no)

Tefera Belachew (teferabelachew@gmail.com)

Gunnar Bjune (g.a.bjune@medisin.uio.no)

Fekadu Abebe (fekadu.abeebe@medisin.uio.no)

**Version:** 3  **Date:** 27 May 2019

**Author’s response to reviews:**

BMC Public Health

Editorial office

We would like to thank the Editorial Office for giving us the chance to resubmit the revised manuscript. Here we have responded for the editors comment.

Editor Comments:

1. Please revise your availability of data and materials statement in your declarations. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.

Response: Thank you for the comment. Now we have revised the availability data statement in the declaration section. The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.